Author's response to reviews

Title: Loss of full length CtBP1 expression enhances the invasive potential of human melanoma

Authors:

Andreas Winklmeier (andreas.wicklmeier@klinik.unir.de)
Ina Poser (ina.poser@web.de)
Keith S Hoek (keith.hoek@usz.ch)
Anja K Bosserhoff (anja.bosserhoff@klinik.uni-regensburg.de)

Version: 5 Date: 6 January 2009

Author's response to reviews:

A conclusion was added and the table 2 was re-laidout.